Author's response to reviews

Title: Clinical decision-making: physicians preferences and experiences

Authors:

Elizabeth Murray (elizabeth.murray@pcps.ucl.ac.uk)
Lance Pollack (lance.pollack@ucsf.edu)
Martha White (mwhite@finance.ucsf.edu)
Bernard Lo (bernie@medicine.ucsf.edu)

Version: 4 Date: 14 March 2007

Author's response to reviews:

1. Job titles have been removed (apologies for missing this last time);
2. Abstract word count removed.
3. "Reference List" has been changed to "References".
4. The table on page 22 is a continuation of the Table on page 21 - it is all one table.
5. The tables have been reformatted.

Please let me know if there is anything further you require.